Author's response to reviews

Title: A randomized controlled trial evaluating the effect of family-based multidisciplinary cognitive behavioral treatment in children with obesity

Authors:

Rimke C Vos (r.vos@hagaziekenhuis.nl)
Jan M Wit (j.m.wit@lumc.nl)
Hanno Pijl (h.pijl@lumc.nl)
Carolien C Kruyff (c.kruyff@hagaziekenhuis.nl)
Euphemia CAM Houdijk (m.houdijk@hagaziekenhuis.nl)

Version: 2 Date: 8 September 2010

Author's response to reviews: see over
Date: 8 September 2010

Dear Editor:

Please find enclosed our the revised manuscript entitled "A randomized controlled trial evaluating the effect of family-based multidisciplinary cognitive behavioral treatment in children with obesity". We would be pleased if you would consider it as a candidate for a study protocol in the BMC Public Health.

This study is approved by the regional ‘Medical Ethical Committee Zuid West Holland’ and a trial registration number is obtained by the ‘International Standard Randomised Controlled Trial Number Register’. All parents and children gave their written informed consent. Copies of the documentation referring to the funding and ethical approval (long version in Dutch and a short translation in English) has been enclosed as additional files.

This manuscript represents our original unpublished work and it is currently not under consideration for review to any other journal. All authors agree for submission and declare no conflict of interest.

On behalf of all authors,

Sincerely,

R. C. Vos
Juliana Children’s Hospital/ HagaHospital
Department of Paediatrics
Sportlaan 600
2566 MJ the Hague
The Netherlands
Phone: 0031(0)70-2107418
Fax: 0031(0)70-2106186
Email: r.vos@hagaziekenhuis.nl